Author's response to reviews

Title: A Bleeding Kiss: Intramural haematoma secondary to balloon angioplasty

Authors:

Cheerag Shirodaria (cshirodaria@doctors.org.uk)
William J van Gaal (william@vangaal.org)
Adrian P Banning (adrian.banning@orh.nhs.uk)

Version: 2 Date: 16 May 2007

Author's response to reviews:

We are grateful to the Reviewers for their helpful comments. We agree that the case report would be enhanced by the addition of a video clip of the case. We have therefore added an intravascular ultrasound movie detailing the intramural haematoma in more detail.

Regards,

Dr Cheerag Shirodaria